\documentstyle{article}
\pagestyle{empty}
\topmargin 0.0in
\headsep 0.0in
\textwidth 6.5in
\textheight 9.0in
\setlength{\parindent}{0.0in}
\oddsidemargin 0.0in
\evensidemargin 0.0in

\begin{document}
%% start public page
\vspace*{-0.5in}
\noindent
{\huge \bf 6.837 Final Project Evaluation, Team \underline{ \hspace{1.0in}}}
\vspace*{0.5in}
\noindent
{\Large Evaluator: \underline{ \hspace{1.5in}}} \hspace{0.2in}
{\LARGE \bf Project Grade (Overall): \underline{ \hspace{0.5in}}}\\

{\Large Team Presentation:}
{\small \bf [Was the detail level appropriate?  Were design and implementation
(in some cases, plans and execution) separated?  Was there good
use of visuals? How were questions handled?  Did the presentation 
convey the technical challenges faced and met?]}

\vspace{2.0in}

{\Large Final Report:}
{\small \bf [Was the report structured appropriately?  Was the problem
well-motivated?  Were technical successes clearly stated, and technical
failures clearly acknowledged?]}

\vspace{2.0in}

{\Large Checkpoints \& Consistency:}
{\small \bf [Did the team submit a reasonable initial proposal
and checkpoints?  Did the team make reasonable progress 
throughout the project period?]}

\vspace{0.5in}

{\Large Project Quality:} 
{\small \bf [What is your overall impression of project quality?]}

\vspace{0.5in}

%% start private page
\newpage
\noindent
{\LARGE \bf This page is for the staff only. Team \underline{ \hspace{1.0in}}}
\vspace*{1.0in}

{\LARGE \bf Point allocations:} 
\vspace*{0.25in}

\begin{tabular}{lr}
{\LARGE Team Presentation} & \underline{ \hspace{1.0in}} / 10 \\
{\LARGE Final Report} & \underline{ \hspace{1.0in}} / 10 \\
{\LARGE Checkpoints \& Consistency} & \underline{ \hspace{1.0in}} / 10 \\
{\LARGE Project Quality} & \underline{ \hspace{1.0in}} / 10 \\
{~} \\
{\LARGE Total} & \underline{ \hspace{1.0in}} / 40 \\
\end{tabular}

\vspace{0.5in}

\noindent
{\LARGE \bf Staff Comments:} \\

\vspace{0.5in}

\end{document}
